Author's response to reviews

Title: FGFR4 Gly388Arg polymorphism contributes to prostate cancer development and progression: A meta-analysis of 2618 cases and 2305 controls

Authors:

Bin Xu (xb15896450810@126.com)
Na Tong (xb15896450810@126.com)
Shu-Qiu Chen (xb15896450810@126.com)
Zheng-Dong Zhang (xb15896450810@126.com)
Zengjun Wang (xb15896450810@126.com)
Ming Chen (xb15896450810@126.com)
Li-Xin Hua (xb15896450810@126.com)

Version: 4 Date: 23 January 2011

Author's response to reviews: see over
Formatting changes to be made:
-------------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name, Initial, Last name' (e.g. Joe F Bloggs)
Response: We have made the changes.

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend.
Response: We have added the titles before Figure legends section.

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Alternatively, please cite the files as Additional file 1 etc., and include an additional files section in the manuscript.
Response: We have included the tables in the manuscript in a tables section following the references.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
Response: We have remove the highlighting.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: We have checked it.

The corresponding email was changed into mingchenseu@gmail.com.